Author's response to reviews

Title: Treatment of scoliosis in patients affected with Prader-Willi syndrome using various different techniques

Authors:

Tiziana Greggi (tiziana.greggi@ior.it)
Kostas Martikos (kostasmartikos@hotmail.com)
Francesco Lolli (francesco.lolli@fastwebnet.it)
Georgios Bakaloudis (bakaloudisgeorgios@hotmail.com)
Mario Di Silvestre (mario.disilvestre@ior.it)
Alfredo Cioni (alfredo.cioni@ior.it)
Giovanni Barbanti Bròdano (giovanni.barbantibrodano@ior.it)
Stefano Giacomini (stefano.giacomini@ior.it)

Version: 6 Date: 14 June 2010

Author's response to reviews:

Concerning the major revision, you say "Figures: The first figure referenced in the manuscript text should be Figure 1. The second, figure 2, etc. Please reorder the figure citation so that they are cited in consecutive order. All figures must be cited; figures not cited will be excluded from the final manuscript. Please do not use ranges when listing figures. Please list all the figures, for example: Figures 5, 6, 7, 8 and 9." As a matter of fact, the first figure referenced in the ms was Figure 1, then the second, Figure 2, and so on until Fig. 4. All figures were already cited, we have added the letters after the numbers (1a, 1b, 1c, 1d and 1e) for all of the figures. We do hope, we have understood correctly what you meant, but are open to make any necessary change.

Regarding our table, we were unable to delete vertical lines and do regret.